Burden of CKD-Associated Pruritus and Adverse Clinical Outcomes in Patients Receiving Dialysis: The Stockholm Creatinine Measurements (SCREAM) Project

Anne-Laure Faucon, Catherine M. Clase, Helena Rydell, Milica Uhde, Peter Barany, Marie Evans,* and Juan-Jesús Carrero*

Rationale & Objective: Pruritus is a common but not well-characterized complaint of patients receiving maintenance dialysis. This study sought to quantify the burden of pruritus and its associated adverse health outcomes in this population.

Study Design: Observational study.

Setting & Participants: All patients receiving maintenance dialysis in Stockholm, Sweden, during 2005-2021.

Exposure: Clinically recognized pruritus defined using *International Classification of Diseases, Tenth Revision* codes or a prescription for antipruritus treatments (including UV therapy).

Outcomes: All-cause mortality, severe infection-related hospitalizations (composite of endocarditis, peritoneal dialysis-related peritonitis, hemodialysis/peritoneal dialysis-related catheter infection, sepsis due to *Staphylococcus* spp., or skin infection) and incident diagnoses of anxiety/ depression and sleep disorders.

Analytical Approach: Multivariable logistic regression and cause-specific hazards models to analyze factors associated with prevalent and new-onset pruritus, respectively. Multivariable cause-specific hazards models with time-varying exposure were used to explore the association of prevalent and new-onset pruritus with adverse health outcomes.

Results: Among 3,281 dialysis recipients (median age, 64 years; 66% men; 69% receiving hemodialysis, 77% with incident dialysis), 456 (14%) had pruritus at enrollment. During a median follow-up of 3.3 (IQR, 1.3-9.2) years, 539 (19%) additional patients experienced pruritus. Older age, female sex, a lower serum albumin level, and higher C-reactive protein, serum and phosphorus calcium, levels were independently associated with pruritus. Compared with patients without pruritus. patients with pruritus were at a higher risk of sleep disorders (adjusted HR, 1.96; 95% Cl, 1.60-2.39), developing anxiety/depression (adjusted HR, 1.56; 95% CI, 1.23-1.98), and being hospitalized for severe infections (adjusted HR, 1.36; 95% Cl, 1.18-1.57), the latter attributed to higher risk of sepsis and peritoneal dialysis-related peritonitis. There was detectable association between no the development of pruritus and all-cause mortality.

Limitations: Potential misclassification bias if pruritus is not clinically recognized, lack of information on pruritus intensity/severity, use of diagnostic codes for exposure and outcome diagnoses.

Conclusions: At least one third of patients experience pruritus during their first years undergoing dialysis, and pruritus was consistently associated with adverse health outcomes.

Visual Abstract online

Complete author and article information provided before references.

Correspondence to A.-L. Faucon (annelaure. faucon@ki.se)

*M.E. and J.-J.C. contributed equally to this work.

Am J Kidney Dis. 85(1):45-54. *Published online July* 26, 2024.

doi: 10.1053/ j.ajkd.2024.05.013

© 2024 The Authors. Published by Elsevier Inc. on behalf of the National Kidney Foundation, Inc. This is an open access article under the CC BY license (http://creativecommons. org/licenses/by/4.0/).

tients who are undergoing dialysis or will be soon.³ Selfreported pruritus has been described in approximately 25% of patients with non–dialysis-dependent CKD stages 3-5⁴ and approximately 40% of patients undergoing maintenance dialysis.^{5,6} The unremitting and refractory nature of this condition may persist over a period of years,^{7,8} and the condition causes distress that ranges from sporadic discomfort to complete restlessness. There is a need to better characterize the burden of this condition and identify populations at risk of developing

Druritus, a strong sensation of itching, is one of the most burdensome uremic symptoms experienced by

patients with advanced chronic kidney disease (CKD)^{1,2}

and is considered an important research priority for pa-

There is a need to better characterize the burden of this condition and identify populations at risk of developing pruritus who might benefit from screening or diagnosis and subsequent initiation of treatment or who might be potential research participants.³ Patients underrecognize and underreport their pruritus, probably because they are unaware of its link with CKD, and may interpret it as a symptom they must live with.^{6,9,10} Further, nephrologists and other health care providers underestimate its prevalence and severity, prioritizing other health issues over pruritus.^{6,9,11} Data from previous observational studies showed that pruritus was associated with poor quality of life, restless sleep, depression, infections, and mortality.4-6.12-17 However, most of the previous work comes from crosssectional studies conducted in patients undergoing hemodialysis (HD) with self-reported symptoms of pruritus, which may lead to possible recall or misclassification bias. Studies involving patients undergoing peritoneal dialysis (PD) are scarce,¹⁶ often from single centers and with small sample sizes. More dynamic evaluations that included





PLAIN-LANGUAGE SUMMARY

Pruritus is a common but not well-characterized symptom of patients receiving dialysis. We analyzed data from 3,281 patients receiving maintenance hemodialysis or peritoneal dialysis in the region of Stockholm, Sweden. At baseline, 14% of patients had pruritus, and pruritus developed in an additional 19% of patients during their time receiving dialysis. We identified conditions associated with the development of pruritus (eg, older age, female sex, lower serum albumin level, and higher C-reactive protein, serum calcium, and phosphorus levels) and observed that the presence of pruritus was associated with higher risks of sleep disorders, developing anxiety and depression, and being hospitalized for severe infections. No association between pruritus and all-cause mortality was identified.

incident cases would allow for a better quantification of the burden of this condition and potential risk factors.

Using a large contemporary health care extraction of patients receiving maintenance dialysis in Stockholm, Sweden, we quantified the population affected by clinically recognized pruritus, their clinical determinants, and the associated adverse health outcomes.

Methods

Population Source and Study Population

The Stockholm Creatinine Measurement (SCREAM) study is a health care use cohort of the complete population of the Stockholm region of Sweden.²³ Via each citizen's unique personal identification number, SCREAM links various national and regional government-run registries that collect information on health care access and health processes. For this study, we included all patients initiating or undergoing maintenance dialysis (HD or PD) during the period of 2005-2021. Information for these patients was obtained primarily from the interlinked Swedish Renal Registry, a nationwide quality registry that annually randomly records clinical and laboratory data of patients receiving maintenance dialysis at any nephrology clinic. Therefore, in this study, incident dialysis recipients are those with less than 1 year of dialysis therapy. These data were enriched with administratively coded clinical diagnoses and procedures (via linkage with the regional health care use database), medications dispensed at any Swedish pharmacies (via linkage with the Drug Prescription Registry), and vital status (via linkage with the Death Registry). The date of the first registered annual visit recorded in the Swedish Renal Registry constituted the index date (ie, baseline) and the start of follow-up (Fig S1). The study was approved by the Swedish Ethical Review Authority. Individual-level informed consent was not required by the National Review Ethics Board for this type of study.

Exposure and Outcomes

This study encompasses 2 complementary analyses. In the first analysis, the study outcome was clinically recognized pruritus, and we assessed its prevalence, incidence, and main baseline determinants. In the second analysis, pruritus was considered as a time-varying exposure, and we analyzed the association between prevalent/new-onset pruritus and a range of adverse health outcomes.

Clinically recognized pruritus was defined by the presence of a clinical diagnosis (International Classification of Diseases, Tenth Revision codes L298, L299, and F458) in the patient medical records, the chronic use of pruritus medications (at least 2 consecutive dispensations), and/or the use of UV therapy (in the absence of a diagnosis of psoriasis; Table S1). At the time of data collection, treatments used by nephrologists in Sweden for the management of CKDassociated pruritus included hydroxyzine, clemastine, gabapentin, pregabalin (the latter 2 being assumed to be prescribed for pruritus treatment in the absence of a clinical diagnosis of diabetes, peripheral neuropathy, or epilepsy), and an extemporaneously compounded topical preparation with levomenthol, camphor, chloral-hydrate (specific for pruritus treatment and classically known as klådsalva in Swedish, meaning "ointment for itching"). Prevalent pruritus was defined by the presence of a clinical diagnosis of pruritus at baseline or a recent (12 months before enrollment) prescription for a pruritus treatment. New-onset pruritus was defined by a de novo clinical diagnosis or an initial prescription for a chronic treatment for pruritus during follow-up.

Adverse health outcomes included all-cause mortality, severe infection-related hospitalizations, de novo anxiety/ depression, and sleep disorders. Severe infection-related hospitalizations considered the composite of endocarditis, PD-related peritonitis, HD-/PD-related catheter infection, sepsis due to Staphylococcus spp. or unspecified Staphylococcus, or skin infection (erysipelas, infective dermatitis, impetigo, abscess, cellulitis, and other skin infections). De novo anxiety/depression or sleep disorders were ascertained in patients free from these conditions at baseline by incident clinical diagnoses or the initiation of related treatments. Detailed definitions of study outcomes are presented in Table S1. For each outcome, patients were followed until the first occurrence of an event, kidney transplant, death, or the end of follow-up (December 31, 2021).

Covariates

Covariates were derived at the index date and were updated at the time of new pruritus onset during the follow-up period. They included age, sex, comorbidities, clinical and laboratory data, information on dialysis therapy, as well as ongoing medications (detailed definitions in Table S1). Comorbidities considered included hypertension, diabetes mellitus, dyslipidemia, ischemic heart disease, myocardial infarction, cerebrovascular disease, peripheral artery disease, heart failure, arrhythmia, anxiety/depression, restless legs syndrome, peripheral neuropathy, and skin infection. Clinical assessment data included systolic and diastolic blood pressure, body mass index, hemoglobin, C-reactive protein (CRP), serum albumin, total calcium, phosphate, and parathyroid hormone. Information on dialysis therapy included dialysis modality and vintage, Kt/V, and type of vascular access. Ongoing medications considered, besides treatments for pruritus, included the use of phosphate binders, calcimimetic agents, iron supplements, erythropoiesis-stimulating agents, and opioids dispensed in the 6 months before the index date.

Statistical Analyses

Values are expressed as means and standard deviations for continuous variables with normal distribution, medians with IQRs for variables with nonnormal distribution, and numbers with percentages of the total for categorical variables.

We evaluated baseline clinical determinants of prevalent pruritus by multivariable logistic regression and of newonset pruritus with a multivariable cause-specific hazards model overall and by dialysis modality. Determinants were selected a priori on the basis of current knowledge on the pathophysiology of pruritus⁷ and included age, sex, diabetes, systolic blood pressure, body mass index, cardiovascular disease, peripheral neuropathy, restless legs syndrome, dialysis modality, dialysis vintage, Kt/V, hemoglobin, CRP, serum calcium, serum phosphate, parathyroid hormone, and serum albumin levels. Continuous variables were standardized as per 1-standard deviation increase. Kt/V for patients undergoing PD was standardized referent to all patients undergoing PD, and HD was treated similarly. When converted to standard deviations, the 2 metrics were combined.

Next, we analyzed the association between pruritus and adverse health outcomes with time-varying cause-specific hazards models. Pruritus was considered a time-varying exposure, thus patients developing new-onset pruritus developed contributed to the nonexposed group from enrollment until the occurrence of pruritus, and contributed to the exposed group thereafter (Fig S2). For the evaluation of risks of de novo anxiety/depression and sleep disorders, we excluded patients with a history of these conditions at baseline. The proportional hazard assumption was checked using log(–log[S]) plots and Schoenfeld residuals against time.

Study covariates had no missing values or were missing in <5% of patients (Table S2) except for parathyroid hormone, serum calcium, and CRP, which were missing in 19%, 26%, and 27% of participants, respectively. Because these laboratory tests are part of the routine monitoring of patients receiving dialysis, we assumed them to be missing at random and attributed this to a lack of reporting to the registry. We performed single imputations, replacing missing data by the next laboratory test carried backward or the median value of the cohort if no other laboratory measurement was reported to the registry.

We performed subgroup analyses to assess the consistency of our results by stratum of age, sex, presence/ absence of cardiovascular disease, diabetes, and dialysis modality. As a sensitivity analysis, we repeated our outcome analysis as follows: (1) excluding patients with pruritus at baseline and evaluating adverse health outcomes after new-onset pruritus (n = 2,825), (2) in patients receiving incident dialysis (n = 2,510), and splitting the pruritus definition into (3) the presence of a clinical diagnosis and (4) prescription of specific treatments (chronic use of hydroxyzine or clemastine or use of the pruritus-specific extemporaneously compounded topical preparation klådsalva), ie, those that do not depend on an assumption regarding the indications for prescription.

We followed the Strengthening the Reporting of Observational Studies in Epidemiology statement for the reporting of observational studies.²⁴ All statistical analyses were conducted using R software (version 3.6.3; R Foundation for Statistical Computing).

Results

Baseline Characteristics and Identification of Pruritus Cases

We identified 3,281 patients receiving maintenance dialysis (77% of whom had been receiving dialysis for less than 1 year at baseline) in Stockholm during 2005-2021. Their median age was 64 years, 66% were men, 69% were receiving HD, and their mean dialysis vintage was 2.2 years (Table 1). At baseline, 456 (13.9%) patients (14.8% of patients receiving HD and 12.0% of patients receiving PD) had clinically recognized pruritus. Overall, 36% of pruritus cases were identified through International Classification of Diseases codes and 64% solely through initiation of a pruritus treatment. Compared with patients who were free from pruritus, patients with prevalent pruritus were older; more often women; had higher prevalences of cardiovascular disease, anxiety/ depression, sleep disorders, and skin infections; and were more often receiving HD.

During a median follow-up of 3.3 (IQR, 1.3-9.2) years, an additional 539 patients (19.1%) who were pruritus-free at baseline (21.2% of patients receiving HD and 14.4% of patients receiving PD) experienced pruritus, with a total period prevalence of 33%. Overall, 19.3% of incident cases were identified through International Classification of Diseases codes and the remaining solely through initiation of a pruritus treatment. This period prevalence was higher in HD than in PD (36.0% of patients receiving HD and 26.4% of patients receiving PD had pruritus; P < 0.01).

Table 1. Characteristics of Patients Receiving Maintenance Dialysis at Enrollment (Baseline) and at the Time of New Pruritus Occurrence After Enrollment

	Pruritus at Enr	ollment		New-Oncet Pruritus After	
Characteristic	No (n = 2,825) Yes (n = 456)		P Value	Enrollment (n = 539)	
Demographic and clinical data					
Median age, y	64 [51-73]	66 [55-75]	0.003	67 [56-75]	
Female sex	944 (33%)	183 (40%)	0.006	200 (37%)	
Systolic BP, mm Hg	140 ± 23	139 ± 24	0.5	140 ± 24	
Diastolic BP, mm Hg	78 ± 14	75 ± 15	0.002	77 ± 15	
BMI. kg/m ²	25.3 ± 5.4	25.6 ± 5.3	0.2	25.6 ± 5.3	
Comorbidities					
Hypertension	2.777 (98%)	456 (100%)	0.009	539 (100%)	
Diabetes mellitus	1.049 (37%)	214 (47%)	< 0.001	203 (38%)	
Dvslipidemia	1.313 (47%)	227 (50%)	0.2	238 (44%)	
Ischemic heart disease	651 (23%)	150 (33%)	< 0.001	156 (29%)	
Myocardial infarction	443 (16%)	103 (23%)	< 0.001	107 (20%)	
Cerebrovascular disease	431 (15%)	75 (16%)	0.6	104 (19%)	
Peripheral arterial disease	313 (11%)	73 (16%)	0.003	88 (16%)	
Heart failure	776 (26%)	158 (35%)	0.002	178 (33%)	
Arrhythmia	559 (20%)	110 (24%)	0.04	127 (24%)	
Peripheral neuropathy	43 (2%)	13 (3%)	0.07	14 (3%)	
Restless legs syndrome	90 (3%)	23 (5%)	0.06	14 (3%)	
Depression/anxiety	478 (17%)	129 (28%)	< 0.001	132 (25%)	
Sleep disorders	965 (34%)	243 (53%)	< 0.001	266 (49%)	
Skin infection	458 (16.2%)	97 (21%)	0.009	127 (24%)	
Biological data		01 (2170)		(, o,	
Hemoglobin g/dl	114+15	114+15	04	115+15	
C-reactive protein mg/l	11.9 + 24.4	167+324	0.002	131+226	
Serum albumin g/l	318+52	311+58	0.005	319+47	
Serum calcium, mmol/L	2.3 + 0.2	2.3 ± 0.2	0.1	2.3 + 0.2	
Serum phosphate, mmol/L	1.7 + 0.5	1.7 + 0.5	0.6	1.8 + 0.6	
PTH_pmol/I	335+452	342+418	0.8	40.4 + 64.3	
Dialysis modality	00.0 ± 10.2	04.2 ± 41.0	0.0	10.1 2 01.0	
Dialysis duration v	22+52	25+54	0.3	38 (54)	
Dialysis modality, PD	907 (32%)	194 (97%)	0.04	128 (24%)	
Access	007 (0270)	124 (2770)	0.04	120 (2470)	
HD catheter	863 (30%)	134 (29%)	_0.1	160 (30%)	
Arteriovenous fistula	780 (28%)	144 (32%)		184 (34%)	
Arteriovenous araft	256 (9%)	51 (11%)		62 (12%)	
Atten access	19 (0 7%)	3 (0.7%)		5 (0.9%)	
PD catheter	907 (32%)	194 (97%)		128 (24%)	
Kt/V per week	007 (0270)	121 (2770)		120 (2170)	
	21+05	21+05	0.3	23+04	
	23+06	22+06	0.0	2.0 ± 0.4	
Orgoing medications	2.5 ± 0.0	2.2 ± 0.0	0.5	2.5 ± 0.7	
	70 (3%)	158 (25%)	<0.001	102 (26%)	
Clomasting	118 (1%)	177 (20%)	<0.001		
Progebolin		1/ (3970)	<0.001		
Cabapantin	25 (0.9%)		<0.001	23 (376) 41 (89/)	
Cappage and tables properties for pruritys (Ulådesha)	40 (2 %)	27 (0%)	<0.001		
Enthronoicei etimulating agente	0 004 (70%)	27 (0%)	<u> </u>	208 (749/)	
	2,224 (19%)		0.9	090 (74%)	
	1,300 (49%)	247 (04%)	1.0	202 (02.0)	
Description bindore	1 700 (610/)	304 (670/)	0.02	365 (68%)	
	200 (110/)		<0.03		
Opiolas	309 (11%)	99 (ZZ%)	~0.00T	92 (1/70)	

Categorical variables are reported as frequencies (percentages), and continuous variables are reported as means ± standard deviations or medians [first and third interquartile values]. Abbreviations: BMI, body mass index; BP, blood pressure; HD, hemodialysis; PD, peritoneal dialysis; PTH, parathyroid hormone.



Figure 1. Determinants of prevalent (odds ratios) and new-onset (hazard ratios) pruritus were estimated using a multivariable logistic regression model and a cause-specific hazards regression model, respectively. All covariates were included in the same model, but the figure has been divided into 2 panels for easier presentation. The risk associated with continuous variables (ie, age, systolic blood pressure, body mass index, dialysis vintage, Kt/V, and all biological data) is expressed per each 1–standard deviation increase. Abbreviations: BMI, body mass index; CI, confidence interval; CRP, C-reactive protein; HR, hazard ratio; OR, odds ratio; PD, peritoneal dialysis; PTH, parathyroid hormone; S-calcium, total serum calcium; S-phosphate, serum phosphate; SBP, systolic blood pressure.

Determinants of Prevalent and New-Onset Pruritus

The clinical determinants of prevalent or new-onset pruritus were consistent in general and included older age, female sex (vs male), higher levels of CRP, serum calcium, and phosphate, and a lower level of serum albumin (Fig 1; Table S3). Diabetes was associated with prevalent but not with new-onset pruritus. PD (vs HD) tended to be inversely associated with pruritus, but the confidence intervals were broad (Fig 1; Table S3). Determinants of pruritus were consistent in general across dialysis modalities (HD vs PD; Figs S3 and S4) and dialysis vintage (incident vs prevalent; Figs S5 and S6), although the smaller sample sizes across subgroups yielded broad confidence intervals.

Impact of Pruritus on Adverse Clinical Outcomes

During follow-up, 1,532 deaths, 949 infection-related hospitalizations, 328 new cases of anxiety/depression, and 485 new cases of sleep disorders occurred. Compared with pruritus-free periods, pruritus was associated with higher risks of de novo anxiety/depression (adjusted hazard ratio, 1.56; 95% confidence interval, 1.23-1.98) and sleep disorders (adjusted hazard ratio, 1.96; 95% confidence interval, 1.60-2.39). The presence of pruritus was also associated with the risk of subsequent severe infections (adjusted hazard ratio, 1.36; 95% confidence interval, 1.18-1.57), mainly attributed to a higher risk of sepsis and PD-related peritonitis (Fig 2; Tables S4 and S5). Pruritus was not associated with the risk of all-cause mortality.

Results were consistent across subgroups of age, sex, diabetes, cardiovascular disease, and dialysis modality (Fig 3).

Excluding patients with prevalent pruritus at enrollment (Table S6) or restricting the study population to patients receiving incident dialysis (Table S7) yielded similar associations with adverse health outcomes. Similar trends were also observed when pruritus was defined only by International Classification of Diseases codes (Table S8) or by the use of treatment (Table S8).

Discussion

We observed that at least one third of patients (33%) experience pruritus during their first years receiving maintenance dialysis. We also identified subpopulations that are consistently at higher risk of having pruritus, including older patients, women, and those with higher levels of serum calcium, phosphate, CRP, or lower levels of serum albumin. Finally, our findings showed that patients with pruritus were at high risk of adverse health outcomes including de novo anxiety/depression, sleep disorders, and infection-related hospitalizations.

Pruritus is common in patients undergoing dialysis. Although data from older studies and clinical experience suggest that the prevalence of pruritus was substantially higher in the early years of dialysis,²⁵ it still remains high, affecting at least one third of the patients receiving maintenance dialysis in our study. Our observed period prevalence is consistent with other previous contemporary reports,^{12,18,20} showing, for instance, that approximately 37% of patients self-reported being moderately to extremely bothered by pruritus in DOPPS (the Dialysis Outcomes and Practices Pattern Study) during 2009-2018.¹² Whereas preceding evidence comes from crosssectional studies that were mainly restricted to patients

Outcomes

Sepsis due to Staph.

PD-related peritonitis

Endocarditis

Catheter-related infection

Skin infection-related hosp.

Pruritus and adverse health outcomes

Outcomes	Pruritus-free, n	Pruritus, n	HR (95%CI)*
All-cause mortality	980	552	0.94 (0.84-1.05)
Infection-related hosp.**	612	337	1.36 (1.18-1.57)
Depression	210	118	1.56 (1.23-1.98)
Sleep disorders	331	154	1.96 (1.60-2.39)

Pruritus-free, n

249

259

62

128

92

Pruritus, n

178

101

44

83

56



Figure 2. Adjusted hazard ratios for the association between pruritus and adverse health outcomes. The figure represents the number of events in each time period (pruritus-free and pruritus periods) and adjusted hazard ratios for the association between pruritus, adverse health outcomes, and severe infection-related hospitalizations. Single asterisk: cause-specific hazards models were fitted using pruritus as a time-varying exposure and were adjusted for age, sex, hypertension, diabetes mellitus, dyslipidemia, ischemic heart disease, heart failure, arrythmia, peripheral artery disease, cerebrovascular disease, peripheral neuropathy, restless legs syndrome, systolic blood pressure, body mass index, hemoglobin, C-reactive protein, serum albumin, serum calcium, parathyroid hormone, dialysis modality, dialysis vintage, Kt/V, iron, erythropoiesis-stimulating agents, calcimimetics, and phosphate binders. For de novo anxiety/depression and sleep disorders, analyses were performed in patients free from these conditions at baseline. Double asterisks: severe infection-related hospitalization, a composite of hospitalizations for sepsis due to Staphylococcus spp. or unspecified Staphylococcus, catheter-related infection, peritoneal dialysis-related peritonitis, skin infection (erysipelas, infective dermatitis, impetigo, abscess, cellulitis, and other skin infections), or endocarditis. Abbreviations: CI, confidence interval; HR, hazard ratio; PD, peritoneal dialysis.

receiving HD, we included patients receiving HD and PD and also evaluated new cases during follow-up to better capture the total pruritus burden of this population. However, because pruritus is widely underreported by patients^{6,9} and underrecognized or underestimated by health care professionals,^{6,11} the real burden is possibly higher and could explain, at least in part, the poor outcomes in this population. Moreover, patient-reported symptom burden may differ from the clinically recognized symptoms we report. A recent study conducted in incident dialysis recipients from the European QUALity (EQUAL) cohort described a self-reported pruritus prevalence of approximately 60%, of which 25% of the patients were bothered at least "quite a bit."²⁶ In this study, pruritus was one of few symptoms that increased in burden during the first year of dialysis.

An important finding of the present study is the consistent association between pruritus and a range of adverse health outcomes. The observational nature of our study cannot demonstrate the causality of this finding and

may simply highlight a population with multiple complications. The higher rate of infection in patients with pruritus can be explained by immune system dysregulation—that results in new-onset pruritus and higher risks of infection—and the presence of a central venous catheter.²⁷ However, it may also be a direct consequence of scratching and excoriation, especially skin infections and sepsis due to Staphylococcus spp. Other studies have also linked pruritus with infection-related hospitalizations, 12,13,16 skin-related infections,¹² catheter-associated infections,¹³ and intravenous antibiotic agent use.¹⁶

The distress caused by persistent pruritus has been reported to impact patients' quality of life, 5,12,14,16,28 and we observed higher subsequent risks of developing sleep disorders. Pruritus may be worse during the night,⁸ and other reports have related pruritus intensity with poor quality of sleep^{5,8,12,14,15,17,29}; restless sleep⁶; feeling tired, "washed out," or "drained"^{12,14}; daily somnolence or difficulty sleeping at night⁵; and longer recovery time from HD sessions.¹² This persistent distress may

	rearry				Severe infect	on-related	nospitalizatio	5115	
Subgroup	n events/N	HR (95%CI)*	p-value	1 .	Subgroup	n events/N	HR (95%CI)*	p-value	
Age < 65 years	447 / 1680	0.99 (0.81-1.22)	0.44	• •	Age < 65 years	411 / 1680	1.30 (1.05-1.60)	0.64	
Age > 65 years	1085 / 1601	0.91 (0.80-1.04)			Age > 65 years	464 / 1601	1.42 (1.17-1.72)		
Women	527 / 1127	1.14 (0.95-1.38)	0.003		Women	302 / 1127	1.34 (1.06-1.70)	0.95	
Men	1005 / 2154	0.83 (0.72-0.96)		-	Men	573/2154	1.36 (1.14-1.62)		
No diabetes	810/2018	0.87 (0.74-1.01)	0.24	-	No diabetes	502 / 2018	1.26 (1.04-1.52)	0.44	
Diabetes	722 / 1263	1.01 (0.86-1.19)		+	Diabetes	373 / 1263	1.43 (1.16-1.78)		
No CVD	447 / 1527	0.83 (0.67-1.02)	0.78	-	No CVD	361 / 1527	1.45 (1.15-1.82)	0.28	
CVD	1085 / 1754	0.98 (0.86-1.12)		- ÷	CVD	514 / 1754	1.30 (1.08-1.55)		-
Hemodialysis	1155 / 2250	0.94 (0.83-1.07)	0.53		Hemodialysis	501 / 2250	1.39 (1.16-1.67)	0.1	
Peritoneal dialysis	377 / 1031	1.05 (0.83-1.32)			Peritoneal dialysis	374 / 1031	1.31 (1.05-1.64)		
				0 1 2 3 Hazard ratio, [95%CI]	-				0 1 2 3 Hazard ratio, [95%CI]
Depression					Sleep disord	lers			
Depression Subgroup	n events/N	HR (95%CI)*	p-value		Sleep disord	lers n events/N	HR (95%CI)*	p-value	
Depression Subgroup Age < 65 years	n events/N 132 / 1313	HR (95%CI)* 1.68 (1.17-2.41)	p-value 0.84		Sleep disord Subgroup Age < 65 years	lers n events/N 209/1112	HR (95%CI)* 2.72 (2.04-3.63)	p-value 0.002	
Depression Subgroup Age < 65 years Age > 65 years	n events/N 132 / 1313 170 / 1331	HR (95%CI)* 1.68 (1.17-2.41) 1.59 (1.16-2.20)	p-value 0.84		Sleep disord Subgroup Age < 65 years Age > 65 years	lers n events/N 209 / 1112 232 / 927	HR (95%Cl)* 2.72 (2.04-3.63) 1.54 (1.15-2.05)	p-value 0.002	
Depression Subgroup Age < 65 years Age > 65 years Women	n events/N 132 / 1313 170 / 1331 119 / 880	HR (95%CI)* 1.68 (1.17-2.41) 1.59 (1.16-2.20) 1.96 (1.34-2.85)	p-value 0.84 . 0.16 .		Sleep disord Subgroup Age < 65 years Age > 65 years Women	lers n events/N 209 / 1112 232 / 927 145 / 693	HR (95%Cl)* 2.72 (2.04-3.63) 1.54 (1.15-2.05) 2.01 (1.42-2.83)	p-value 0.002 0.66	
Age < 65 years Age < 65 years Age > 65 years Women Men	n events/N 132/1313 170/1331 119/880 183/1764	HR (95%Cl)* 1.68 (1.17-2.41) 1.59 (1.16-2.20) 1.96 (1.34-2.85) 1.31 (0.96-1.80)	p-value 0.84 0.16		Sleep disord Subgroup Age < 65 years Age > 65 years Women Men	lers n events/N 209 / 1112 232 / 927 145 / 693 296 / 1346	HR (95%Cl)* 2.72 (2.04-3.63) 1.54 (1.15-2.05) 2.01 (1.42-2.83) 1.90 (1.48-2.45)	p-value 0.002 0.66	
Depression bgroup Age < 65 years	n events/N 132/1313 170/1331 119/880 183/1764 171/1683	HR (95%Cl)* 1.68 (1.17-2.41) 1.59 (1.16-2.20) 1.96 (1.34-2.85) 1.31 (0.96-1.80) 1.75 (1.28-2.41)	p-value 0.84 0.16 0.34		Sleep disord subgroup Age < 65 years Age > 65 years Women Men No diabetes	lers n events/N 209 / 1112 232 / 927 145 / 693 296 / 1346 261 / 1325	HR (95%Cl)* 2.72 (2.04-3.63) 1.54 (1.15-2.05) 2.01 (1.42-2.83) 1.90 (1.48-2.45) 2.15 (1.64-2.81)	p-value 0.002 0.66	
Depression Age < 65 years	n events/N 132/1313 170/1331 119/880 183/1764 171/1683 131/961	HR (95%CI)* 1.68 (1.17-2.41) 1.59 (1.16-2.20) 1.96 (1.34-2.85) 1.31 (0.96-1.80) 1.75 (1.28-2.41) 1.39 (0.96-2.01)	p-value 0.84 0.16 0.34		Sleep disord subgroup Age < 65 years	lers n events/N 209/1112 232/927 145/693 296/1346 261/1325 180/714	HR (95%CI)* 2.72 (2.04-3.63) 1.54 (1.15-2.05) 2.01 (1.42-2.83) 1.90 (1.48-2.45) 2.15 (1.64-2.81) 1.65 (1.20-2.27)	p-value 0.002 0.66 0.26	
Depression Age < 65 years	n events/N 132/1313 170/1331 119/880 183/1764 171/1683 131/961 118/1260	HR (95%C)* 1.68 (1.17-2.41) 1.59 (1.16-2.20) 1.96 (1.34-2.85) 1.31 (0.96-1.80) 1.75 (1.28-2.41) 1.39 (0.96-2.01) 1.52 (1.02-2.27)	p-value 0.84 0.16 0.34 0.34 0.82		Sleep disord Subgroup Age < 65 years	ers nevents/N 209/1112 232/927 145/693 296/1346 261/1325 180/714 203/1069	HR (95%C)* 2.72 (2.04-3.63) 1.54 (1.15-2.05) 2.01 (1.42-2.83) 1.90 (1.48-2.45) 2.15 (1.64-2.81) 1.65 (1.20-2.27) 1.67 (1.21-2.30)	p-value 0.002 0.66 0.26 0.93	
Depression Age < 65 years	n events/N 132 / 1313 170 / 1331 119 / 880 183 / 1764 171 / 1683 131 / 961 118 / 1260 184 / 1384	HR (95%CI)* 1.68 (1.17-2.41) 1.59 (1.16-2.20) 1.96 (1.34-2.85) 1.31 (0.96-1.80) 1.75 (1.28-2.41) 1.39 (0.96-2.01) 1.52 (1.02-2.27) 1.55 (1.15-2.11)	p-value 0.84 0.16 0.34 0.34 0.34		Sleep disord Subgroup Age < 65 years	Bers N 209/1112 232/927 145/693 296/1346 296/1346 296/1346 261/1325 180/714 203/1069 238/970	HR (95%C)* 2.72 (2.04-3.63) 1.54 (1.15-2.05) 2.01 (1.42-2.83) 1.90 (1.48-2.45) 2.15 (1.64-2.81) 1.65 (1.20-2.77) 1.67 (1.21-2.30) 2.14 (1.63-2.80)	p-value 0.002 0.66 0.26 0.93	
Depression Age < 65 years Age > 65 years Women Mon Mon diabetes Diabetes CVD Hemodialysis	n events/N 132 / 1313 170 / 1331 119 / 880 183 / 1764 171 / 1683 131 / 961 118 / 1260 184 / 1384 227 / 1773	HR (95%C)* 1.68 (1.17-2.41) 1.59 (1.16-2.20) 1.96 (1.34-2.85) 1.31 (0.96-1.80) 1.75 (1.28-2.41) 1.39 (0.96-2.01) 1.52 (1.02-2.27) 1.55 (1.15-2.11) 1.64 (1.25-2.16)	p-value 0.84 0.16 0.34 0.34 0.34 0.82		Sleep disord subgroup Age < 65 years Age > 65 years Women Men No diabetes Diabetes No CVD CVD Hemodialysis	Bers N 209/1112 232/927 145/693 296/1346 206/1346 203/1069 203/1069 203/1069 203/970 308/1322	HR (95%C)* 2.72 (2.04-3.63) 1.54 (1.15-2.05) 2.01 (1.42-2.83) 1.80 (1.48-2.45) 2.15 (1.64-2.81) 1.65 (1.20-2.77) 1.67 (1.21-2.30) 2.14 (1.63-2.80) 1.92 (1.51-2.43)	p-value 0.002 0.66 0.26 0.93	
bepression Subgroup Age < 65 years Vomen Men No diabetes Diabetes CVD Hemodialysis	n events/N 132 / 1313 170 / 1331 119 / 880 183 / 1764 171 / 1683 131 / 961 118 / 1260 184 / 1384 227 / 1773 75 / 871	HR (95%C)* 1.68 (1.17-241) 1.59 (1.16-220) 1.96 (1.34-2.85) 1.31 (0.96-1.80) 1.75 (1.28-241) 1.39 (0.96-2.01) 1.52 (1.02-227) 1.55 (1.15-2.11) 1.64 (1.25-2.16) 1.21 (0.72-2.04)	p-value 0.84 0.16 0.34 0.382 0.44		Sleep disord Jage < 65 years	Bers N 209/1112 232/927 145/693 296/1346 261/1325 180/714 203/1069 238/970 308/1322 133/717	HR (95%C)* 2.72 (2.04-3.63) 1.54 (1.15-2.05) 2.01 (1.42-2.83) 1.90 (1.48-2.45) 2.15 (1.64-2.81) 1.65 (1.20-2.27) 1.67 (1.21-2.30) 2.14 (1.63-2.80) 1.92 (1.51-2.43) 2.42 (1.65-3.55)	p-value 0.002 0.66 0.26 0.93 0.93	

Figure 3. Hazard ratios for the association between pruritus and adverse outcomes according to baseline subgroup. Asterisk: cause-specific hazards models were fitted using pruritus as a time-varying exposure and were adjusted for age, sex, hypertension, diabetes mellitus, dyslipidemia, ischemic heart disease, heart failure, arrythmia, peripheral artery disease, cerebrovascular disease, peripheral neuropathy, restless legs syndrome, systolic blood pressure, body mass index, hemoglobin, C-reactive protein, serum albumin, serum calcium, parathyroid hormone, dialysis modality, dialysis vintage, Kt/V, iron, erythropoiesis-stimulating agents, calcimimetics, and phosphate binders. For de novo anxiety/depression and sleep disorders, analyses were performed in patients free from these conditions at baseline. The association between pruritus and the risk of death was stronger in women (P = 0.003 for interaction). The association between pruritus and the risk of sleep disorders was stronger in patients younger than 65 years (P = 0.002 for interaction). The other tested interactions were not significant. Abbreviations: CI, confidence interval; CVD, cardiovascular disease; HR, hazard ratio.

psychologically impact the patient with a higher risk of anxiety/depression, as observed in our study and expanding on previous works.^{5,12,17} Finally, we did not observe any association between pruritus and all-cause mortality, which contrasts with reports from DOPPS^{5,12,14} and may be explained by differences in study power or pruritus ascertainment (self-reported symptoms vs clinically recognized diagnoses).

The pathophysiology of CKD-associated pruritus is not fully understood and is thought to occur in the setting of a complex metabolic environment that may involve uremic toxin deposition, inflammation and/or immune system dysregulation, peripheral neuropathy, and opioid imbalance.⁷ Our multivariable risk analyses of prevalent and new-onset pruritus based on the aforementioned pathophysiological mechanisms is broadly consistent with prior findings. Characteristics and findings of the main epidemiological studies in this regard are summarized in Table S9. In general, the consistency of some clinical predictors is conflicting, possibly impacted by the evaluation of small single-center cohorts of low-powered, cross-sectional analyses and self-reports.

In the present study, pruritus was independently associated with older age and female sex. This agrees with some, ^{6,8,13,14,16,28,30} but not all, ^{5,8,14,16,22,28,29} preceding studies. Although itch-related mechanisms may differ between sexes, ³¹ it is possible that these differences may also be explained by gender differences in the perception and experience of pruritus: men and women may experience and react differently to pruritus and may choose to express it differently; men more often perceive themselves as better able to cope with the physical aspects of their disease or are less prone to tell others about their pruritus, ³² and health care professionals may therefore believe pruritus to be more burdensome and more often require treatment among women.

Pruritus might be related to skin deposition of pruritogens such as calcium, phosphorus, protein-bound uremic toxins, and middle molecules that are less effectively removed by the current dialysis modalities. Our study agrees with this hypothesis, as patients at risk of pruritus had higher calcium and phosphate levels, which is consistent with some, 5,14,16,21,29,33 but not all, 6 prior studies. However, in agreement with a recent study from DOPPS,⁶ we did not find an association between pruritus and parathyroid hormone levels. Also in line with previous studies, our observed association of pruritus with high CRP^{6,34,35} levels of and low levels of albumin^{5,6,14,16,19,30,36} is consistent with the hypothesis that pruritus is a consequence of local microinflammation, systemic inflammatory syndrome, and immune system dysregulation.^{7,35} Neuropathy and dysautonomia may also be involved in the pathophysiology of pruritus,⁷ but, even though the prevalences of restless legs syndrome and peripheral neuropathy tended to be higher in our patients with pruritus, we found no statistically significant association in our study. We acknowledge that these associations may be confounded by the inclusion of gabapentinoids, which are also used in the treatment of restless legs syndrome, in our definition of pruritus. Nevertheless, these conditions are not always well captured by clinical diagnoses. Finally, we did not find an association between hemoglobin level and pruritus. Discrepancies exist in the literature in regard to the impact of anemia and ironrelated parameters on pruritus.^{5,6,8,16}

Dialysis vintage tended to be longer (although the 95%) confidence intervals overlapped) in patients with pruritus than in those without, which was consistent with previous studies^{14-17,34} However, our study population's short dialysis vintage (most patients had less than 1 year of dialysis therapy) compared with previous reports may have contributed to this lack of association. In addition, in line with recent literature,^{6,14,16,17,29,33} we did not observe any association between pruritus and dialysis dose (ie, Kt/V). The direction of the association was the opposite of our expectation, which may be the result of reverse causation bias: a higher prescribed dialysis dose may be the consequence of itching, and there is a natural delay between the reporting of symptoms and the diagnosis and initiation of treatments. A higher prescribed Kt/ V may also be related to lower residual kidney function, which induces a decrease in tubular secretion of proteinbound organic anions and uremic toxins, contributing to pruritus.¹⁸ Other hypotheses are that Kt/V does not accurately quantify the removal of the medium- and largesized molecules involved in the pathophysiology of pruritus or that the range of Kt/V values in our study is narrow and close to the recommended target.

We found that pruritus tended to be less common in patients undergoing PD versus HD, but it was similarly associated with adverse health outcomes. We believe this is important to emphasize because new treatments for pruritus are not yet available for patients undergoing PD. More preserved residual kidney function and less inflammation may explain the lower incidence of pruritus observed in PD. Previous literature has been mixed in this regard: although an equal number of studies report a higher prevalence of pruritus in patients receiving HD^{20,21} or PD,^{19,22} others found a similar prevalence between the 2 dialysis modalities.¹⁸

Strengths of our study include a large sample size of wellphenotyped patients with a relatively high proportion of patients receiving PD, long follow-up, careful study design with evaluation of new cases during follow-up to better capture the pruritus burden in this population, and the setting involving patients from a country with universal taxfunded health care, which minimizes selection bias from disparate access to health care. Limitations include the observational nature of the study, which is prone to residual time-varying confounding; potential misclassification bias if pruritus is not reported or recognized by patients and/or health care professionals or because medications for pruritus are nonspecific, which obliged us to make some considerations and assumptions when ascertaining that medications were used for the management of pruritus and not for other indications; and the lack of assessment of pruritus intensity/ severity-the pruritus status having been extracted from medical records—and the relationship between pruritus and the type of vascular access (central venous catheter vs fistula) as a result of the small number of patients in each subgroup. We also recognize the absence of validation of our approach to identify pruritus. However, the fact that pruritus, by our definition, is predicted by many of the variables we would hypothesize to be predictive, and is associated with the expected adverse outcomes provides construct validity to our definition.

In conclusion, our study illustrates the commonness of clinically recognized pruritus in patients receiving dialysis, identifies populations at risk of having/developing pruritus, and shows that patients with pruritus are at high risk of adverse health outcomes. Our findings may inform clinical decisions for periodic pruritus screening, especially in high-risk populations. Individualized clinical management to prevent and treat this burdensome uremic symptom might significantly improve patients' quality of life and potentially health outcomes. Approaches to pruritus management include optimization of dialysis, treatment of CKD-associated mineral and bone disease, extemporaneous compounded topical preparations, systemic pharmacological treatments, UV therapy,³⁷ and, in HD, difelikefalin, a recently introduced selective agonist of κ-opioid receptors.³⁸

Supplementary Material

Supplementary File (PDF) Figure S1: Study design. Figure S2: Pruritus as a time-varying exposure. **Figure S3:** Associations of baseline variables with prevalent (odds ratios) and new-onset (hazard ratios) pruritus in patients undergoing maintenance hemodialysis.

Figure S4: Associations of baseline variables with prevalent (odds ratios) and new-onset (hazard ratios) pruritus in patients undergoing maintenance peritoneal dialysis.

Figure S5: Associations of baseline variables with prevalent (odds ratios) and new-onset (hazard ratios) pruritus in incident dialysis recipients.

Figure S6: Associations of baseline variables with prevalent (odds ratios) and new-onset (hazard ratios) pruritus in prevalent dialysis recipients.

Table S1: Definitions of comorbidities, medications, and outcomes.

Table S2: Number of missing values at baseline and at the time of new-onset pruritus by time period.

Table S3: Crude and multivariable regression models of baseline determinants of prevalent (odds ratios with 95% Cls) and new-onset (hazard ratios with 95% Cls) pruritus.

Table S4: Incidence rates and hazard ratios for the association between pruritus and adverse clinical outcomes.

 Table S5: Incidence rates and hazard ratios for the association

 between pruritus and severe infection-related hospitalizations.

 Table S6: Hazard ratios for the association between pruritus and adverse clinical outcomes in patients free from pruritus at baseline.

Table S7: Hazard ratios for the association between pruritus and adverse clinical outcomes in incident dialysis recipients.

Table S8: Hazard ratios for the association between pruritus and severe infection-related hospitalizations using different pruritus definitions.

Table S9: Comparison of the main epidemiological studies on pruritus conducted in patients undergoing maintenance dialysis.

Article Information

Authors' Full Names and Academic Degrees: Anne-Laure Faucon, MD, PhD, Catherine M. Clase, MB, MSc, Helena Rydell, MD, PhD, Milica Uhde, PhD, Peter Barany, MD, PhD, Marie Evans, MD, PhD, and Juan-Jesús Carrero, PharmD, PhD.

Authors' Affiliations: Department of Medical Epidemiology and Biostatistics (A-LF, J-JC) and Division of Nephrology, Department of Clinical Science, Intervention and Technology (HR, PB, ME), Karolinska Institutet; Division of Nephrology, Department of Clinical Sciences, Danderyd Hospital, Karolinska Institutet (J-JC); and CSL Vifor (MU), Stockholm, Sweden; Department of Clinical Epidemiology, Institut National de la Santé et de la Recherche Médicale Unité Mixte de Recherche 1018, Centre for Epidemiology and Population Health, Paris-Saclay University, Paris, France (A-LF); and Department of Medicine and Department of Health Research Methods, Evidence, and Impact, McMaster University, Hamilton, ON, Canada (CMC).

Address for Correspondence: Anne-Laure Faucon, MD, PhD, Institutionen för Medicinsk Epidemiologi och Biostatistik, Karolinska Institutet, Nobels väg 12A, 171 77 Stockholm, Sweden. Email: annelaure.faucon@ki.se

Authors' Contributions: Study design: A-LF, ME, J-JC; data acquisition: ME; statistical analyses: A-LF; interpretation: A-LF, ME, J-JC, CC, HR, PB, MU. Each author contributed important intellectual content during manuscript drafting or revision and agrees to be personally accountable for the individual's own contributions and to ensure that questions pertaining to the accuracy or integrity of any portion of the work, even one in which the author was not directly involved, are appropriately investigated

Support: Karolinska Institutet received support from Vifor Pharma to conduct this study. Dr Carrero acknowledges support from the Swedish Research Council and the Swedish Heart and Lung Foundation. Funders had no role in study design, data collection, analysis, reporting, or the decision to submit for publication.

Financial Disclosure: Dr Evans reports personal honoraria for lectures by Astra-Zeneca, Astellas Pharma, Vifor Pharma, Fresenius Healthcare, and Baxter Healthcare and being a member of advisory boards for Astellas, Astra-Zeneca, and Vifor Pharma. Dr Carrero reports funding to Karolinska Institutet by AstraZeneca, Astellas, Amgen, Vifor Pharma, and NovoNordisk; personal honoraria for lectures by Fresenius Kabi, Baxter Healthcare, and Abbott; and being a member of advisory boards for Astellas, AstraZeneca, and GSK. Dr Uhde is an employee of Vifor Pharma. Dr Clase reports that she has received consultation, advisory board membership, honoraria, or research funding from the Ontario Ministry of Health, Sanofi, Pfizer, Leo Pharma, Astellas, Janssen, Amgen, Boehringer-Ingelheim, Baxter, and, through LiV Academy, AstraZeneca. She is Editor-in-Chief of the Canadian Journal of Kidney Health and Disease (CJKHD), the official journal of the Canadian Society of Nephrology. The remaining authors declare that they have no relevant financial interests.

Prior Presentation: Presented in part at the American Society of Nephrology 2023 Kidney Week; November 3, 2023; Philadelphia, PA.

Peer Review: Received October 7, 2023. Evaluated by 2 external peer reviewers, with direct editorial input from a Statistics/ Methods Editor, an Associate Editor, and the Editor-in-Chief. Accepted in revised form May 13, 2024.

References

- Rhee EP, Guallar E, Hwang S, et al. Prevalence and persistence of uremic symptoms in incident dialysis patients. *Kid-ney360*. 2020;1(2):86-92. doi:10.34067/kid.0000072019
- Murtagh FEM, Addington-Hall JM, Edmonds PM, et al. Symptoms in advanced renal disease: a cross-sectional survey of symptom prevalence in stage 5 chronic kidney disease managed without dialysis. J Palliat Med. 2007;10(6):1266-1276. doi:10.1089/jpm.2007.0017
- Manns B, Hemmelgarn B, Lillie E, et al. Setting research priorities for patients on or nearing dialysis. *Clin J Am Soc Nephrol.* 2014;9(10):1813-1821. doi:10.2215/CJN. 01610214
- Sukul N, Speyer E, Tu C, et al. Pruritus and patient reported outcomes in non-dialysis CKD. *Clin J Am Soc Nephrol.* 2019;14(5):673-681. doi:10.2215/CJN.09600818
- Pisoni RL, Wikström B, Elder SJ, et al. Pruritus in haemodialysis patients: International results from the Dialysis Outcomes and Practice Patterns Study (DOPPS). *Nephrol Dial Transplant*. 2006;21(12):3495-3505. doi:10.1093/ndt/gfl461
- Rayner HC, Larkina M, Wang M, et al. International comparisons of prevalence, awareness, and treatment of pruritus in people on hemodialysis. *Clin J Am Soc Nephrol.* 2017;12(12): 2000-2007. doi:10.2215/CJN.03280317
- Verduzco HA, Shirazian S. CKD-associated pruritus: new insights into diagnosis, pathogenesis, and management. *Kidney Int Rep.* 2020;5(9):1387-1402. doi:10.1016/j.ekir.2020.04. 027
- Mathur VS, Lindberg J, Germain M, et al. A longitudinal study of uremic pruritus in hemodialysis patients. *Clin J Am Soc Nephrol.* 2010;5(8):1410-1419. doi:10.2215/CJN.00100110

AJKD

- Aresi G, Rayner HC, Hassan L, et al. Reasons for underreporting of uremic pruritus in people with chronic kidney disease: a qualitative study. *J Pain Symptom Manage*. 2019;58(4):578-586.e2. doi:10.1016/j.jpainsymman.2019.06. 010
- 10. Prasad B, Gagarinova M, Sharma A. Five things to know about pruritus in patients on dialysis. *Can J Kidney Health Dis.* 2023;10:20543581221149620. doi:10.1177/ 20543581221149620
- Weisshaar E, Matterne U, Mettang T. How do nephrologists in haemodialysis units consider the symptom of itch? Results of a survey in Germany. *Nephrol Dial Transplant*. 2009;24(4):1328-1330. doi:10.1093/ndt/gfn769
- Sukul N, Karaboyas A, Csomor PA, et al. Self-reported pruritus and clinical, dialysis-related, and patient-reported outcomes in hemodialysis patients. *Kidney Med.* 2021;3(1):42-53.e1. doi: 10.1016/j.xkme.2020.08.011
- Ting SW, Fan PC, Lin YS, et al. Uremic pruritus and long-term morbidities in the dialysis population. *PLoS One*. 2020;15(10): e0241088. doi:10.1371/journal.pone.0241088
- Kimata N, Fuller DS, Saito A, et al. Pruritus in hemodialysis patients: results from the Japanese Dialysis Outcomes and Practice Patterns Study (JDOPPS). *Hemodial Int.* 2014;18(3): 657-667. doi:10.1111/hdi.12158
- Weiss M, Mettang T, Tschulena U, Passlick-Deetjen J, Weisshaar E. Prevalence of chronic itch and associated factors in haemodialysis patients: a representative cross-sectional study. Acta Derm Venereol. 2015;95(7):816-821. doi:10. 2340/00015555-2087
- Ramakrishnan K, Bond TC, Claxton A, et al. Clinical characteristics and outcomes of end-stage renal disease patients with self-reported pruritus symptoms. *Int J Nephrol Renovasc Dis.* 2013;7:1-12. doi:10.2147/IJNRD.S52985
- Li J, Guo Q, Lin J, Yi C, Yang X, Yu X. Prevalence and associated factors of uraemic pruritus in continuous ambulatory peritoneal dialysis patients. *Intern Med.* 2015;54(22):2827-2833. doi:10.2169/internalmedicine.54.4516
- Lengton R, van der Willik EM, de Rooij ENM, et al. Effect of residual kidney function and dialysis adequacy on chronic pruritus in dialysis patients. *Nephrol Dial Transplant*. 2023;38(6):1508-1518. doi:10.1093/ndt/gfac341
- Min JW, Kim SH, Kim YO, et al. Comparison of uremic pruritus between patients undergoing hemodialysis and peritoneal dialysis. *Kidney Res Clin Pract*. 2016;35(2):107-113. doi:10. 1016/j.krcp.2016.02.002
- Wu HY, Peng YS, Chen HY, et al. A comparison of uremic pruritus in patients receiving peritoneal dialysis and hemodialysis. *Medicine (Baltimore)*. 2016;95(9):e2935. doi:10.1097/ MD.000000000002935
- Hu T, Wang B, Liao X, Wang S. Clinical features and risk factors of pruritus in patients with chronic renal failure. *Exp Ther Med.* 2019;18(2):964-971. doi:10.3892/etm.2019.7588
- Mistik S, Utas S, Ferahbas A, et al. An epidemiology study of patients with uremic pruritus. J Eur Acad Dermatol Venereol. 2006;20(6):672-678. doi:10.1111/j.1468-3083.2006.01570.x
- Carrero JJ, Elinder CG. The Stockholm CREAtinine Measurements (SCREAM) project: fostering improvements in chronic kidney disease care. *J Intern Med.* 2022;291(3):254-268. doi: 10.1111/joim.13418
- 24. von Elm E, Altman DG, Egger M, Pocock SJ, Gøtzsche PC, Vandenbroucke JP. The Strengthening the Reporting of

Observational Studies in Epidemiology (STROBE) statement: guidelines for reporting observational studies. *Lancet.* 2007;370(9596):1453-1457. doi:10.1016/S0140-6736(07) 61602-X

- Young AW, Sweeney EW, David DS, et al. Dermatologic evaluation of pruritus in patients on hemodialysis. N Y State J Med. 1973;73(22):2670-2674.
- de Rooij ENM, Meuleman Y, de Fijter JW, et al. Symptom burden before and after dialysis initiation in older patients. *Clin J Am Soc Nephrol.* 2022;17(12):1719-1729. doi:10.2215/ CJN.09190822
- Ravani P, Palmer SC, Oliver MJ, et al. Associations between hemodialysis access type and clinical outcomes: a systematic review. *J Am Soc Nephrol.* 2013;24(3):465-473. doi:10.1681/ ASN.2012070643
- Rehman IU, Chan KG, Munib S, Lee LH, Khan TM. The association between CKD-associated pruritus and quality of life in patients undergoing hemodialysis in Pakistan: a STROBE complaint cross-sectional study. *Medicine (Baltimore)*. 2019;98(36):e16812. doi:10.1097/MD.000000000016812
- Narita I, Alchi B, Omori K, et al. Etiology and prognostic significance of severe uremic pruritus in chronic hemodialysis patients. *Kidney Int.* 2006;69(9):1626-1632. doi:10.1038/sj.ki. 5000251
- Lopes GB, Nogueira FCP, de Souza MR, et al. Assessment of the psychological burden associated with pruritus in hemodialysis patients using the kidney disease quality of life short form. *Qual Life Res.* 2012;21(4):603-612. doi:10.1007/s11136-011-9964-x
- Martin CE, Clotet-Freixas S, Farragher JF, Hundemer GL. Have we just scratched the surface? A narrative review of uremic pruritus in 2020. *Can J Kidney Health Dis.* 2020;7: 2054358120954024. doi:10.1177/2054358120954024
- Hecking M, Hödlmoser S, Ahmed SB, Carrero JJ. The other way around: living with chronic kidney disease from the perspective of men. *Semin Nephrol.* 2022;42(2):122-128. doi: 10.1016/j.semnephrol.2022.04.003
- Weisshaar E, Weiss M, Passlick-Deetjen J, Tschulena U, Maleki K, Mettang T. Laboratory and dialysis characteristics in hemodialysis patients suffering from chronic itch – results from a representative cross-sectional study. *BMC Nephrol.* 2015;16:184. doi:10.1186/s12882-015-0177-3
- Wu HY, Huang JW, Tsai WC, et al. Prognostic importance and determinants of uremic pruritus in patients receiving peritoneal dialysis: a prospective cohort study. *PLoS One*. 2018;13(9): e0203474. doi:10.1371/journal.pone.0203474
- Kimmel M, Alscher DM, Dunst R, et al. The role of microinflammation in the pathogenesis of uraemic pruritus in haemodialysis patients. *Nephrol Dial Transplant*. 2006;21(3):749-755. doi:10.1093/ndt/gfi204
- Virga G, Visentin I, La Milia V, Bonadonna A. Inflammation and pruritus in haemodialysis patients. *Nephrol Dial Transplant*. 2002;17(12):2164-2169. doi:10.1093/ndt/17.12.2164
- Lipman ZM, Paramasivam V, Yosipovitch G, Germain MJ. Clinical management of chronic kidney disease-associated pruritus: current treatment options and future approaches. *Clin Kidney* J. 2021;14(suppl 3):i16-i22. doi:10.1093/ckj/sfab167
- Fishbane S, Jamal A, Munera C, Wen W, Menzaghi F; KALM-1 Trial investigators. A phase 3 trial of difelikefalin in hemodialysis patients with pruritus. N Engl J Med. 2020;382(3):222-232. doi:10.1056/NEJMoa1912770

